Author’s response to reviews

Title: Association between De Ritis ratio (aspartate aminotransferase/alanine aminotransferase) and oncological outcomes in bladder cancer patients after radical cystectomy

Authors:

Yun-Sok Ha (yunsokha@gmail.com)
Sang Won Kim (mydecision@naver.com)
So Young Chun (soyachun99@naver.com)
Jae-Wook Chung (jeus119@hanmail.net)
Seock Hwan Choi: (skhwan@gmail.com)
Jun Nyung Lee (ljnlover@gmail.com)
Bum Soo Kim (dock97@hanmail.net)
Hyun Tae Kim (urologistk@knu.ac.kr)
Eun Sang Yoo (uroyoo@knu.ac.kr)
Tae Gyun Kwon (tgkwon@knu.ac.kr)
Won Tae Kim (wtkimuro@chungbuk.ac.kr)
Wun-Jae Kim (wjkim@chungbuk.ac.kr)
Tae-Hwan Kim (doctork@knu.ac.kr)

Version: 3 Date: 21 Dec 2018

Author’s response to reviews:

1. We note that the current submission contains some textual overlap with other previously published works, in particular:

"PFKFB4 as a prognostic marker in non-muscle-invasive bladder cancer" (2012)
https://doi.org/10.1016/j.urolonc.2010.08.018
Overlap exists page 14 lines 27-40.

While we understand that you may wish to express some of the same ideas contained in these publications, please be aware that we cannot condone the use of text from previously published work. Please re-phrase these sections to minimise overlap.
Answer> We revised all sentences you indicated. Please consider some terminology overlapping inevitably used. Please check the page 9, 14, and 15 in this revised manuscript.

2. Please include an Acknowledgements section in your Declarations where you acknowledge anyone who contributed towards the article who does not meet the criteria for authorship including anyone who provided professional writing services or materials. Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements section If there are no acknowledgements, please state “Not applicable”.
Answer> We have commented in the acknowledgements section as following; “Not applicable”. Please see the page 21.

3. Please change the title of your ‘Patients and Methods’ section to ‘Methods’.
Answer> We changed it. Please see the page 8.

4. Please represent authors' names using their full initials, not their full/surname, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.
Answer> We corrected them. Please see the page 20-21.

5. Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.
Answer> We added a statement that all authors have read and approved the final manuscript. Please see the page 20-21.

6. Please remove the revision letter document from the end of your manuscript file as this is no longer needed in the publication process.
Answer> We deleted the revision letter.

7. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
Answer> Thank you. We can understand.